Author's response to reviews

Title: CON-COUR study. Interferential therapy in the treatment of chronic constipation in adults: study protocol for a randomized controlled trial

Authors:

Véronique Vitton (Veronique.VITTON@ap-hm.fr)
Alban Benezech (Alban.BENEZECH@ap-hm.fr)
Stéphane Honoré (stephane.honore@ap-hm.fr)
Patrick Sudour (patrick.sudour@ap-hm.fr)
Nathalie Lesavre (nathalie.lesavre@ap-hm.fr)
Pascal Auquier (pascal.auquier@univ-amu.fr)
Karine Baumstarck (karine.baumstarck@univ-amu.fr)

Version: 4
Date: 12 May 2015

Author's response to reviews:

1) The title was changed as requested (CON-COUR study. Interferential therapy in the treatment of chronic constipation in adults: study protocol for a randomized controlled trial)
2) The word count was suppressed (title page)